Author’s response to reviews

Title: Fatigue, Barriers to Physical Activity and Predictors for Motivation to Exercise in Advanced Cancer Patients

Authors:

Johanna Frikkel (Johanna.frikkel@uk-essen.de)
Miriam Goette (Miriam.goette@uk-essen.de)
Mingo Beckmann (Mingo.beckmann@lvr.de)
Stefan Kasper (Stefan.kasper-virchow@uk-essen.de)
Joerg Hense (joerg.hense@uk-essen.de)
Martin Teufel (Martin.teufel@uni-due.de)
Martin Schuler (martin.schuler@uk-essen.de)
Mitra Tewes (mitra.tewes@uk-essen.de)

Version: 3 Date: 03 Mar 2020

Author’s response to reviews:

Dear Mr. LeBlanc, dear Mr. Jagoe,

We again thank you very much for revising our manuscript.

We understand your concerns that the terminology we chose to explain the conducted regression analyses might confuse some reader. We accepted your wording and used it in the corresponding sections of our results (regression analyses for PA: page 6, line 28-33; regression analyses for motivation for PA: page 6, line 35-42).

Furthermore some formal changes, required by the editor, were done.

We hope, that the revision in its current version can be provided for publication in the widely known BMC Palliative Care.

Sincerely
Mitra Tewes